Author's response to reviews

Title: Inhibition of DNA methyltransferase activity and expression by treatment with the pan-deacetylase inhibitor panobinostat in hepatocellular carcinoma cell lines

Authors:

Steffen Zopf (steffen.zopf@uk-erlangen.de)
Matthias Ocker (ocker@staff.uni-marburg.de)
Daniel Neureiter (d.neureiter@salk.at)
Beate Alinger (b.alinger@salk.at)
Susanne Gahr (susanne.gahr@t-online.de)
Markus F. Neurath (markus.neurath@uk-erlangen.de)
Pietro Di Fazio (difazio@med.uni-marburg.de)

Version: 4 Date: 9 August 2012

Author's response to reviews: see over
Cover letter

Dear Editors of BMC Cancer,

We appreciate the comments of the reviewers and thank you for the opportunity to revise our manuscript “Inhibition of DNA methyltransferase activity and expression by treatment with the pan-deacetylase inhibitor panobinostat in hepatocellular carcinoma cell lines” accordingly.

Reviewer 2, Huanjie Shao and Editorial Board:

1. We deleted figure 6 and revised the manuscript accordingly.